Author's response to reviews

Title: Medical theses as part of the scientific training in basic medical and dental education: Experiences from Finland

Authors:

Pentti Nieminen (pentti.nieminen@oulu.fi)
Kirsi Sipila (kirsi.sipila@oulu.fi)
Hanna-Mari Takkinen (hanna-mari.takkinen@oulu.fi)
Marjo Renko (marjo.renko@oulu.fi)
Leila Risteli (leila.risteli@oulu.fi)

Version: 8 Date: 5 December 2007

Author's response to reviews:

Further formatting changes have been completed as requested.

Yours sincerely

Pentti Nieminen